Author’s response to reviews

Title: Cross-sectional associations between residential socio-environmental exposures and cardiovascular diseases

Authors:

Antony Chum (antony.chum@utoronto.ca)
Patricia O’Campo (pat.ocampo@utoronto.ca)

Version: 2 Date: 5 September 2014

Author’s response to reviews:

1) Cover letter was removed from the manuscript
2) Title page with post address was added
3) Line numbering was added to the manuscript
4) Background and Conclusions were sectioned off with the proper titles.
5) Competing interests and Authors’ contributions sections was moved after the Conclusions.